Author's response to reviews

Title: Severe gastric variceal haemorrhage due to splenic artery thrombosis and consecutive arterial bypass

Authors:

Klaus T von Trotha (kvontrotha@ukaachen.de)
Nick Butz (nbutz@ukaachen.de)
Marcel Binnebösel (mbinneboesel@ukaachen.de)
Son Truong (struong@ukaachen.de)
Florian F Behrendt (fbehrendt@ukaachen.de)
Hermann E Wasmuth (hwasmuth@ukaachen.de)
Ulf P Neumann (uneumann@ukaachen.de)
Marc Jansen (mjansen@ukaachen.de)

Version: 4 Date: 17 January 2011

Author's response to reviews:

Ref.: 1938887921447171

Severe gastric variceal haemorrhage due to splenic artery thrombosis and consecutive arterial bypass

Dear Rachel Neilan,

Thank you very much for the opportunity to resubmit our manuscript to BMC Surgery. We are grateful for this opportunity. We revised our manuscript according to the reviewers’ comments. Changes are highlighted in the text.

The title was changed accordingly to the suggestions made by reviewer 1. Also an additional figure showing endoscopic duplex sonography prior to operation was added.

The reference list was completed by two case reports as requested by reviewer 2, namely:


Those additions to the reference list have also been highlighted.
Prof. Dr. M. Jansen has changed hospitals and is now head of Helios Hospital in Berlin, therefore the affiliation within the cover page had to be changed accordingly.

The manuscript was revised by a native English speaker.

Sincerely yours,

Klaus-Thilo von Trotha

Correspondence to:
Klaus-Thilo von Trotha, MD
Department of Surgery, RWTH Aachen University Hospital
Pauwelsstr. 30
52074 Aachen, Germany
Phone: +49-241-80-89500
Fax: +49-241-80-82417
Email: kvontrotha@ukaachen.de